Answering Reviewers

Name of Journal: World Journal of Clinical Cases
Manuscript NO: 69750
Manuscript Type: ORIGINAL ARTICLE

SPECIFIC COMMENTS TO AUTHORS
This article is a good analysis of the relationship between malignant neoplasms of various organs. This manuscript uses a lot of statistical material in the SEER dataset. The authors described an interesting phenomenon of prolongation of the survival time of patients with gastric cancer with previous malignant neoplasms. The authors analyzed the causes of gastric cancer, mutation genes TTN, TP53, MUC16, ARID1A, LRP1B and CNTNAP2.

Answer:
Thank you so much for your comments. We really appreciate you taking the time out to review our article. Thanks again!